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Abstract 

The intermittency of renewable energy sources can be tackled by integration with 

energy storage systems (ESS), which should be economical, efficient and suitable 

for large-scale energy storage. Vanadium redox flow battery (VRFB) has been 

emerging as one of the best ESS for its integration with renewable energy sources 

such as solar and wind power. Nevertheless, for an effective market penetration of 

any new technology, the capital cost is an extremely important factor. VRFB suffers 

from high capital costs primarily due to the high cost of vanadium electrolyte, 

arising from an expensive vanadium precursor (vanadyl sulfate, VOSO4) used 

conventionally. In the first study, vanadium pentoxide (V2O5) is explored as an 

alternate vanadium precursor primarily due to its low-cost and common availability. 

However, the solubility of V2O5 is very low in H2SO4 solutions (~0.4 M VO2+ in 3 

M H2SO4 at 40oC), which makes it impractical for preparing vanadium electrolytes 

of desired concentration (1.5 – 1.6 M Vn+ in 4 M total sulfate). To tackle this 

challenge, a novel electrochemical synthesis route is developed which results in 

synthesis of desired vanadium electrolytes from V2O5. The synthesized electrolytes 

show similar performance to that of conventionally used VOSO4 based electrolyte. 

Moreover, the economic analysis suggests more than 95% reduction in the cost of 

electrolyte synthesis using the developed process over conventional process, which 

would ultimately bring down the capital cost of kW-MW VRFB installations. 

Nevertheless, the availability of V2O5 could pose a serious threat to the supply-

chain for VRFB applications since more than 95% of global annual production of 

V2O5 is utilized in metallurgical and catalyst industries. Furthermore, the 

production of V2O5 cannot be scaled-up since it is a co-product of iron and titanium 

mining from V-bearing titanoferrous magnetite ores (primary sources). In the 
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second study, waste vanadium  sources such as spent vanadium catalysts (SVC) are 

explored as secondary vanadium sources for synthesizing vanadium electrolytes. 

However, the presence of variety of impurities (Na, K, Al, Fe, Cs, Mg, Ti, Ca, Zn, 

and Ni) makes it impractical for synthesizing vanadium electrolytes. To tackle this 

challenge, a novel vanadium extraction process is developed which resulted in 

extraction of direct-use V2O4 and V2O3 vanadium precursors. The electrolytes 

synthesized from these extracted vanadium powders result in similar performance 

to that of commercial electrolytes. Further, economic analysis is done to evaluate 

the commercial feasibility of the developed process. The cost of electrolyte 

synthesis using extraction route is around 2.73 $/L, which is significantly lower 

than reported V2O5 routes (~5.27 $/L). Lastly, a multi-cell VRFB system is designed 

and developed based on plate and frame design, where shunt currents are found to 

be a major performance loss parameter. The link between high shunt currents and 

associated bipolar plate degradation is established successfully, which is not 

reported anywhere in the literature. This novel finding is used to optimize the design 

and performance of a multi-cell VRFB system.  

Keywords: Electrolyte; Extraction; Flow battery; Multi-cell stack; Spent vanadium 

catalyst; Shunt current; Vanadium.  
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सारांश 
नवीकरणीय ऊजार् स्रोतों की अंतरिवरोिधता को ऊजार् भंडारण प्रणािलयों (ESS) के साथ एकीकृत 

करके हल िकया जा सकता ह,ै जो आिथर् क, कुशल और बड़े पैमान ेपर ऊजार् भंडारण के िलए 

उपयुक्त होनी चािहए। वैनािडयम रडेॉक्स फ्लो बैटरी (VRFB) नवीकरणीय ऊजार् स्रोतों जैस ेिक 

सौर और पवन ऊजार् के साथ एकीकरण के िलए सबस ेउपयुक्त ESS के रूप में उभर कर सामन े

आई ह।ै हालांिक, िकसी भी नई तकनीकी के प्रभावी बाजार में प्रवेश के िलए पूंजी लागत एक 

अत्यंत महत्वपूणर् कारक होती ह।ै VRFB उच्च पूंजी लागत से ग्रस्त ह,ै जो मुख्य रूप से वैनािडयम 

इलेक्ट्रोलाइट की उच्च लागत के कारण होती ह,ै जो परपंरागत रूप से महगं ेवैनािडयम अग्रणी 

(वैनाडाइल सले्फट, VOSO4) स े उत्पन्न होती ह।ै पहल ेअध्ययन में, वैनािडयम पेंटॉक्साइड 

(V2O5) को एक वैकिल्पक वैनािडयम अग्रणी के रूप में अन्वेिषत िकया गया ह,ै मुख्य रूप से 

इसकी कम लागत और सामान्य उपलब्धता के कारण। हालांिक, V2O5 की घुलनशीलता H2SO4 

समाधानों में बहुत कम ह ै(~0.4 M VO2+ 3 M H2SO4 में 40 °C पर), िजसस ेयह वांिछत सांद्रता 

(1.5 – 1.6 M Vn+ 4 M कुल सले्फट में) के वैनािडयम इलेक्ट्रोलाइट्स तैयार करन े के िलए 

अव्यावहािरक बन जाता ह।ै इस चुनौती से िनपटन े के िलए, एक नवीनतम इलेक्ट्रोकेिमकल 

संश्लेषण मागर् िवकिसत िकया गया ह,ै जो V2O5 स े वांिछत वैनािडयम इलेक्ट्रोलाइट्स का 

संश्लेषण करता ह।ै संश्लेिषत इलेक्ट्रोलाइट्स परपंरागत VOSO4 आधािरत इलेक्ट्रोलाइट्स के 

समान प्रदशर्न िदखात े हैं। इसके अितिरक्त, आिथर् क िवश्लेषण से यह संकेत िमलता ह ै िक 

िवकिसत प्रिक्रया के माध्यम स ेइलेक्ट्रोलाइट संश्लेषण की लागत में परपंरागत प्रिक्रया की तुलना 

में 95% स ेअिधक की कमी आएगी, िजसस ेkW-MW VRFB प्रितष्ठानों की पूंजी लागत में 

कमी आएगी। िफर भी, V2O5 की उपलब्धता VRFB अनुप्रयोगों के िलए आपूितर्  श्रृंखला के िलए 

एक गंभीर खतरा उत्पन्न कर सकती ह,ै क्योंिक िवश्व उत्पादन का 95% से अिधक V2O5 धातुकमर् 

और उत्प्रेरक उद्योगों में उपयोग होता ह।ै इसके अलावा, V2O5 का उत्पादन बढ़ाया नहीं जा सकता 

ह ैक्योंिक यह लौह और टाइटेिनयम खनन स ेउत्पन्न एक सह-उत्पाद ह।ै दूसर ेअध्ययन में, व्यथर् 

वैनािडयम स्रोतों जैस े िक व्यथर् वैनािडयम उत्प्रेरकों (SVC) का अन्वेषण िकया गया ह,ै जो 

वैनािडयम इलेक्ट्रोलाइट्स के संश्लेषण के िलए िद्वतीयक वैनािडयम स्रोत के रूप में काम आ 
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सकत ेहैं। हालांिक, िविभन्न अशुिद्धयों (Na, K, Al, Fe, Cs, Mg, Ti, Ca, Zn और Ni) की 

उपिस्थित इस ेवैनािडयम इलेक्ट्रोलाइट्स के संश्लेषण के िलए अव्यावहािरक बना देती ह।ै इस 

चुनौती स ेिनपटन ेके िलए, एक नवीन वैनािडयम िनष्कषर्ण प्रिक्रया िवकिसत की गई ह,ै िजसके 

पिरणामस्वरूप सीध ेउपयोग के िलए V2O4 और V2O3 वैनािडयम अग्रणी का िनष्कषर्ण हुआ। 

इन िनष्किषर् त वैनािडयम पाउडरों स ेसंश्लेिषत इलेक्ट्रोलाइट्स वािणिज्यक इलेक्ट्रोलाइट्स के 

समान प्रदशर्न करत े हैं। इसके अलावा, िवकिसत प्रिक्रया की व्यावसाियक व्यवहायर्ता का 

मूल्यांकन करन े के िलए आिथर् क िवश्लेषण िकया गया ह।ै िनष्कषर्ण मागर् का उपयोग करके 

इलेक्ट्रोलाइट संश्लेषण की लागत लगभग 2.73 डॉलर/लीटर ह,ै जो V2O5 मागोर्ं (~5.27 

डॉलर/लीटर) की तुलना में काफी कम ह।ै अंत में, एक बहु-कोिशका VRFB प्रणाली को प्लेट 

और फे्रम िडजाइन के आधार पर िडज़ाइन और िवकिसत िकया गया ह,ै जहा ंशंट करेंट्स (shunt 

currents) को एक प्रमुख प्रदशर्न हािन कारक के रूप में पाया गया ह।ै उच्च शंट करेंट्स और 

संबंिधत बाइपोलर प्लेट के क्षय के बीच िलंक को सफलतापूवर्क स्थािपत िकया गया ह,ै जो पहल े

कहीं भी िरपोटर् नहीं िकया गया था। इस नवीनतम खोज का उपयोग करके बहु-कोिशका VRFB 

प्रणाली के िडजाइन और प्रदशर्न को अनुकूिलत िकया गया ह।ै 

मुख्य शब्द: इलेक्ट्रोलाइट; िनष्कषर्ण; फ्लो बैटरी; बहु-कोिशका सै्टक; व्यथर् वैनािडयम उत्प्रेरक; 

शंट करेंट्स; वैनािडयम। 
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